\begin{table}%t3
%\centering
\par
  \caption{\label{tab: classif} Classification of objects based on radio, optical and X-ray properties.}
 \begin{tabular}{l c c c c c c r r r} 
   \hline \hline
   \noalign{\smallskip}
    Name & $R$\tablefootmark{a}  & R. ref. & O\tablefootmark{b}& O. ref. & X-ray\tablefootmark{c} & X-ray ref. & Note & Core Cl. & Dyn. Cl.\\
    \hline
    Abell 4038   	  & H & Sl98, Sl01   &  ? & Bu04      & NCC        & Sn06           & Y & INT & NOM \\  
    Abell 2199  	  & -- &             &  ? & Oe01      & CC         & Jo02           & N & CC  & NOM \\  
    2A 0335+096		  & -- &             &  -- &           & CC         & Ma03           & N & CC  & NOM \\ 
    Abell 2052  	  & -- &             &  N & Di87, Ma92 & CC         & Bl03           & N & CC  & NOM \\  
    Abell 576   	  & N &  Fe07       &  -- &           & MRG \& NCC & Mo96, Ke04, Du07 & N & NCC & MRG \\ %tab A576
    Abell 3571  	  & N &  Ve02       &  --  &           & NCC        & Oh06, Sn06, Da08 & Y & INT & NOM \\ %tab A3571
    Abell 119   	  & N &  Mu04       &  --  &           & MRG \& NCC & Sr06, Ro06      & N & NCC & MRG \\ %tab A119
    MKW 03		  & -- &             &  -- &           & CC         & Ma02, Su07      & N & CC  & NOM \\ %tab MKW3s
    Abell 3376            & R &  Ba06       &  N & Gi97      & MRG \& NCC & Ro06, Ca09      & N & NCC & MRG \\ %tab A3376
    Abell 1644  	  & -- &             &  N & Tu01      & INT        & Re04, Du05      & N & INT & NOM \\ %tab A1644
    Abell 4059  	  & -- &             &  -- &           & CC         & Ch04, Ca09      & N & CC  & NOM \\ %tab A4059
    Abell 3558  	  & N &  Ba98       &  Y & Ba98      & INT        & Ro07           & N & INT & NOM \\ %tab A3558
    Abell 3562  	  & H &  Gi05       &  Y & Ba98      & MRG \& NCC & Fi04           & N & NCC & MRG \\ %tab A3562
    Triangulum Austr.     & -- &             &  -- &           & NCC        & Ch07           & N & NCC & NOM \\ %tab TRIANG_AUS
    Hydra A		  & -- &             &  -- &           & CC         & Da01           & N & CC  & NOM \\ %tab hydra
    Abell 754             & H & Ka01, Ba03   &  ? & Ro98      & MRG \& NCC & Kr03, He04      & N & NCC & MRG \\ %tab A754
    Abell 85    	  & -- &             &  N & Ma92, Br09 & MRG \& CC  & Ke02, Dr05      & N & CC  & MRG \\ %tab A85
    Abell 2319  	  & H & Fe97, Ca06   &  Y & Oe95      & MRG \& NCC & Oh04           & N & NCC & MRG \\ %tab A2319
    Abell 3158  	  & -- &             &  Y & Jo08      & MRG \& NCC & Sn08, Gh09      & N & NCC & MRG \\ %tab A3158
    Abell 1795  	  & -- &             &  -- &           & CC         & Et02           & N & CC  & NOM \\ 
    Abell 399   	  & N &   Gi06      &  Y & Gi97      & MRG \& NCC & Sa04, Ro06      & N & NCC & MRG \\ %tab A399
    Abell 401   	  & H &   Ba03      &  Y & Gi97      & MRG \& NCC & Sa04, Bo08      & N & NCC & MRG \\ %tab A401
    Abell 3112  	  & -- &             &  -- &           & CC         & Ta03           & N & CC  & NOM \\ %tab A3112
    Abell 2029  	  & -- &             &  -- &           & CC         & Cl04           & N & CC  & NOM \\ %tab A2029
    Abell 2255  	  & H &  Pi08       &  -- &           & MRG \& NCC & Sa06           & N & NCC & MRG \\ %tab A2255
    Abell 1650  	  & -- &             &  -- &           & CC         & Do05, Ca09      & Y & INT & NOM \\ %tab A1650
    Abell 2597  	  & -- &             &  -- &           & CC         & Mo05, Ca09      & N & CC  & NOM \\ %tab A2597
    Abell S0084  	  & -- &             &  -- &           & INT        & Si09           & N & INT & NOM \\ %tab AS0084
    Abell 2034  	  & -- &             &  -- &           & MRG \& NCC & Ke03, Bl07      & N & NCC & MRG \\ %tab A2034
    Abell 2051	          & -- &             &  -- &           & NCC        & Pr07, Cr08      & N & NCC & NOM \\ %tab A2051
    Abell 3814	          & -- &             &  -- &           & CC         & Cr08, Le08      & N & CC  & NOM \\ %tab A3814
    Abell 2050	          & -- &             &  -- &           & NCC        & Pr07, Cr08      & N & NCC & NOM \\ %tab A2050
    RXCJ1141.4-1216	  & -- &             &  -- &           & CC         & Pr07, Cr08      & N & CC  & NOM \\ %tab RXCJ1141.4-12
    Abell 1084	          & -- &             &  -- &           & CC         & Pr07, Cr08      & N & CC  & NOM \\ %tab A1084
    Abell 1068  	  & -- &             &  -- &           & CC         & Wi04           & N & CC  & NOM \\ %tab A1068
    Abell 3856            & -- &             &  -- &           & NCC        & Pr07, Cr08      & N & NCC & NOM \\ %tab A3856
    Abell 3378	          & -- &             &  -- &           & CC         & Pr07, Cr08      & N & CC  & NOM \\ %tab A3378
    Abell 0022	          & -- &             &  -- &           & NCC        & Pr07, Cr08      & N & NCC & NOM \\ %tab A0022
    Abell 1413  	  & -- &             &  -- &           & INT        & Vi05, Bl07, Ca09 & N & INT & NOM \\ %tab A1413
    Abell 2328	          & -- &             &  -- &           & NCC        & Pr07, Cr08      & N & NCC & NOM \\ %tab A2328
    Abell 3364	          & -- &             &  -- &           & NCC        & Pr07, Cr08      & N & NCC & NOM \\ %tab A3364
    Abell 2204  	  & -- &             & -- &           & CC         & Sa09           & N & CC  & NOM \\ %tab A2204
    Abell 0907	          & - &             &  -- &           & CC         & Vi05, Cr08      & N & CC  & NOM \\ %tab A907
    Abell 3888	          & -- &             &  -- &           & NCC        & Cr08           & N & NCC & NOM \\ %tab A3888
    RXCJ2014.8-2430	  & -- &             &  -- &           & CC         & Cr08           & N & CC  & NOM \\ %tab RXCJ2014.8-24
    Abell 3404	          & -- &             &  -- &           & INT        & Cr08           & N & INT & NOM \\ %tab A3404
    Abell 1914  	  & H & Ba03        &  -- &           & MRG \& NCC & Go04, Bl07      & N & NCC & MRG \\ %tab A1914
    Abell 2218  	  & H & Gi00        &  Y &  Gi97     & MRG \& NCC & Go04, Bl07      & N & NCC & MRG \\ %tab A2218
    Abell 1689  	  & -- &             &  ? & Gi01, Lo06 & MRG \& CC  & Pe98, An04, Ca09 & Y & INT & NOM \\ %tab A1689
    Abell 383   	  & -- &             &  -- &           & CC         & Vi05, Ca09      & N & CC  & NOM \\ %tab A383
    Abell 115N   	  & ? &  Go01       &  Y & Ba07b     & MRG \& CC  & Gu05           & N & CC  & MRG \\ %tab A115N
    Abell 2163  	  & H &  Fe01       &  Y & Ma08      & MRG \& NCC & Go04           & N & NCC & MRG \\ %tab A2163
    Abell 963   	  & N &  Ca08       &  -- &           & NCC        & Sm05, Bl07, Ca09 & Y & INT & NOM \\ %tab A963
    Abell 209   	  & H &  Ve07       &  Y & Da02, Me04 & NCC        & Ca09           & N & NCC & MRG \\ %tab A209
    Abell 773   	  & H &  Go01       &  Y & Ba07a     & MRG \& NCC & Go04, Ca09      & N & NCC & MRG \\ %tab A773
    \hline
    Abell 1763  	  & N &  Ve08       &  Y & Fa08      & MRG \& NCC & Du08, Ca09      & N & NCC & MRG \\ %tab A1763
    Abell 2390  	  & M &  Ba03       &  N & Le91      & CC         & Vi05, Ca09      & N & CC  & NOM \\ %tab A2390
    Abell 2667  	  & N &  Ve08       &  -- &           & CC         & Co06, Ca09      & N & CC  & NOM \\ %tab A2667
    RX~J2129.3+0005	  & -- &             &  -- &           & CC         & Ca09           & N & CC  & NOM \\ 
    \hline
  \end{tabular}
\par
 \tablefoot{\tablefoottext{a}{Radio emission;} 
    \tablefoottext{b}{optical substructure;}
    \tablefoottext{c} {X-ray classification; a detailed description
    of the table is provided in Sect.~\ref{sec: class sch}.}}
    \tablebib{References  divided by band of electromagnetic spectrum.\\
  {References for radio observations}: (Sl98) \citet{Sl98}; (Sl01) \citet{Sl01}; (Fe07) \citet{Fe07};
  (Ve02) \citet{Ve02}; (Mu04) \citet{Mu04}; (Ba06) \citet{Ba06}; (Ba98) \citet{Ba98};
  (Gi05) \citet{Gi05}; (Ka01) \citet{Ka01}; (Ba03) \citet{Ba03}; (Fe97) \citet{Fe97};
  (Ca06) \citet{Ca06}; (Gi06) \citet{Gi06}; (Pi08) \citet{Pi08}; (Gi00) \citet{Gi00};
  (Go01) \citet{Go01}; (Fe01) \citet{Fe01}; (Ca08) \citet{Ca08}; (Ve07) \citet{Ve07};
  (Ve08) \citet{Ve08}.\\
%
  {References for optical observations}: (Bu04) \citet{Bu04}; (Oe01) \citet{Oe01}; (Di87) \citet{Di87};
  (Ma92) \citet{Ma92}; (Br09) \citet{Br09}; (Gi97) \citet{Gi97}; (Tu01) \citet{Tu01}; (Ro98) \citet{Ro98};
  (Oe95) \citet{Oe95};
  (Jo08) \citet{Jo08}; (Gi01) \citet{Gi01}; (Lo06) \citet{Lo06}; (Ba07b) \citet{Ba07b}; (Ma08) \citet{Ma08};
  (Ba07a) \citet{Ba07a}; (Fa08) \citet{Fa08}; (Le91) \citet{Le91}.\\
%
  {References for X-ray observations}: (Sa06) \citet{Sa06}; (Jo02) \citet{Jo02}; (Ma03) \citet{Ma03};
  (Bl03) \citet{Bl03}; (Mo96) \citet{Mo96}; (Ke04) \citet{Ke04}; (Du07) \citet{Du07}; (Oh06) \citet{Oh06};
  (Sn06) \citet{Sn06}; (Da08) \citet{Da08}; (Sr06) \citet{Sr06}; (Ro06) \citet{Ro06};
  (Ma02) \citet{mazzotta02}; (Su07) \citet{Su07}; (Re04) \citet{Re04}; (Du05) \citet{Du05};
  (Ch04) \citet{Ch04};  (Ro07) \citet{rossetti07}; (Fi04) \citet{Fi04}; (Ch07) \citet{Ch07};
  (Da01) \citet{Da01}; (Kr03) \citet{Kr03}; (He04) \citet{He04}; (Ke02) \citet{Ke02}; (Dr05) \citet{Dr05};
  (Oh04) \citet{Oh04}; (Sn08) \citet{snowden08}; (Gh09) \citet{Gh09}; (Et02) \citet{Et02};
  (Sa04) \citet{Sa04}; (Bo08) \citet{bourdin08};
  (Ta03) \citet{Ta03}; (Cl04) \citet{Cl04};
  (Sa06) \citet{Sa06}; (Do05) \citet{Do05}; (Mo05) \citet{Mo05}; (Si09) \citet{Si09};
  (Ke03) \citet{Ke03}; (Cr08) \citet{Cr08}; (Le08) \citet{leccardi08b}; (Pr07) \citet{pratt07};
  (Wi04) \citet{Wi04}; (Vi05) \citet{vikh05}; (Bl07) \citet{baldi07}; (Go04) \citet{Go04};
  (Pe98) \citet{peres98}; (An04) \citet{An04}; (Gu05) \citet{Gu05}; (Sm05) \citet{Sm05};
  (Da02) \citet{Da02}; (Me04) \citet{Me04}; (Du08) \citet{Du08}; (Co06) \citet{Co06};
  (Ca09) \citet{Ca09}.}
  \end{table}